Author's response to reviews

Title: Empiric guideline-recommended weight-based vancomycin dosing and mortality in methicillin-resistant Staphylococcus aureus bacteremia: a retrospective cohort study

Authors:

Ronald G Hall II (ronald.hall@ttuhsc.edu)
Christopher A Giuliano (ek2397@wayne.edu)
Krystal K Haase (krystal.haase@ttuhsc.edu)
Kathleen A Thompson (khazlewo@uwyo.edu)
Christopher R Frei (freic@uthscsa.edu)
Nicolas A Forcade (nforcade8@gmail.com)
Sara D Brouse (sara.brouse@uky.edu)
Todd Bell (Todd.Bell@ttuhsc.edu)
Roger J Bedimo (roger.bedimo@va.gov)
Carlos A Alvarez (carlos.alvarez@ttuhsc.edu)

Version: 3 Date: 21 March 2012

Author's response to reviews:

We have added the names of the bodies that reviewed the study at each site (North Texas Veterans Health Care System, Texas Tech University Health Sciences Center, and University of Texas Health Science Center, San Anotnio) to the methods section of the manuscript (page 5, Study location and patients).